\begin{table}%t1
\caption{\label{tb:moldata-ls}New molecular data for \coc{12}$^{a}$.}
\par
\small%\centerline
$$ {
{
\begin{tabular}{cc r@{.}l r@{.}l r@{ }l c r@{.}l c r@{.}l r@{.}l r@{.}l r@{.}l r@{.}l ll}
\hline\hline
%&&&&&&&&&&&&&& \\[-8pt]
Band$^{{b}}$ & ER90$^{{b}}$ & \mc{$\lambda_0$} & \mc{$\nu_0$} & \mc{ID} & $v'$ & \mc{$\osc$} & $A_\el{tot}$ & \mc{$\eta$} & \mc{$\bvu$} & \mc{$\dvu$} & \mc{$\weu$} & \mc{$\wexeu$} & References & Notes \\
\#             & \#          & \mc{(\AA)} & \mc{(\pcm)} & \mc{ }                      &   & \mc{ }      & (\ps)        & \mc{ }   & \mc{(\pcm)} & \mc{(\pcm)} & \mc{(\pcm)} & \mc{(\pcm)} & & \\
\hline
\phantom{A0}1A & \phantom{A0}7A &  912&7037 & 109~564&6 & &$^1\Pi$ & 0 & 3&4(--3) & 1(10) & 1&00 & 1&9226 & 5&9(--5) & 2169&8 & 13&29 & 1 & c \\
\phantom{B0}1B & \phantom{B0}7B &  913&4002 & 109~481&0 & $(5p\sigma)$&$^1\Sigma^+$ & 1 & 1&7(--3) & 9(10) & 1&00 & 1&829 & 1&0(--5) & 2214&2 & 15&15 & 1,2 & d \\
\phantom{C0}1C & \phantom{C0}7C &  913&4255 & 109~478&0 & $(5p\pi)$&$^1\Pi$ & 1 & 1&7(--3) & 1(10) & 1&00 & 1&9568 & 1&0(--4) & 2214&2 & 15&15 & & d \\
\phantom{D0}1D & \phantom{D0}7D &  913&6678 & 109~449&0 & &$^1\Sigma^+$ & 2 & 2&7(--2) & 9(10) & 1&00 & 1&7781 & 5&4(--5) & 2169&8 & 13&29 & 1, 2, 3 & c \\
\phantom{A0}2A & \phantom{A0}8A &  915&7258 & 109~203&0 & $(6p\pi)$&$^1\Pi$ & 0 & 2&0(--3) & 1(11) & 1&00 & 1&5796 & 6&737(--6) & 1562&8 & 13&93 & & e, f \\
\phantom{B0}2B & \phantom{B0}8B &  915&9708 & 109~173&8 & $(6p\sigma)$&$^1\Sigma^+$ & 0 & 7&9(--3) & 1(11) & 1&00 & 1&685 & 1&0(--4) & 2214&2 & 15&15 & 3 & d \\
\phantom{0}3 & \phantom{A0}9A &  917&2721 & 109~018&9 & &$^1\Pi$ & 2 & 2&3(--2) & 5(11) & 1&00 & 1&671 & 7&2(--5) & 2169&8 & 13&29 & 2 & c \\
\phantom{0}4 & \phantom{B0}9B &  919&2097 & 108~789&1 & $(6s\sigma)$&$^1\Sigma^+$ & 0 & 2&8(--3) & 1(11) & 1&00 & 2&1357 & 4&633(--5) & 2214&2 & 15&15 & 4, 5 & d \\
\phantom{0}5 & \phantom{C0}9C &  920&1410 & 108~679&0 & $I'(5s\sigma)$&$^1\Sigma^+$ & 1 & 2&8(--3) & 1(11) & 1&00 & 1&9050 & 6&0(--6) & 2291&2 & 0&0 & & g \\
\phantom{0}6 & 10 &  922&7561 & 108~371&0 & $(5d\sigma)$&$^1\Sigma^+$ & 0 & 6&3(--3) & 3(11) & 1&00 & 1&9675 & 6&318(--6) & 2214&2 & 15&15 & & d, h \\
\phantom{0}7 & 11 &  924&6309 & 108~151&3 & &$^1\Sigma^+$ & 1 & 5&2(--3) & 1(11) & 1&00 & 1&8669 & 4&0(--5) & 2169&8 & 13&29 & & c \\
\phantom{0}8 & 12 &  925&8093 & 108~013&6 & $W(3s\sigma)$&$^1\Pi$ & 3 & 2&0(--2) & 4(11) & 1&00 & 1&6506 & 1&1(--4) & 1745&1 & $-4$&22 & 6 & g, i \\
\phantom{0}9 & 13 &  928&6575 & 107~682&3 & &$^1\Pi$ & 2 & 6&7(--3) & 4(10) & 1&00 & 1&936 & 3&1(--5) & 2169&8 & 13&29 & 1, 2, 3 & c \\
10 & 14 &  930&0611 & 107~519&8 & &$^1\Pi$ & 2 & 6&3(--3) & 1(11) & 1&00 & 1&8202 & 2&6(--5) & 2169&8 & 13&29 & & c \\
11 & \phantom{A}15A &  931&0744 & 107~402&8 & &$^1\Pi$ & 0 & 6&0(--3) & 1(11) & 1&00 & 1&650 & 1&0(--5) & 2169&8 & 13&29 & & c \\
12 & \phantom{B}15B &  931&6547 & 107~335&9 & $(5p\pi)$&$^1\Pi$ & 0 & 1&2(--2) & 3(11) & 1&00 & 1&8682 & 4&3(--5) & 2214&2 & 15&15 & 2 & d, j \\
13 & \phantom{C}15C &  933&0583 & 107~174&4 & $(5p\sigma)$&$^1\Sigma^+$ & 0 & 2&2(--2) & 3(10) & 1&00 & 2&128 & 1&0(--5) & 2214&2 & 15&15 & 1, 2 & d, i \\
14 & 16 &  935&6638 & 106~876&0 & &$^1\Sigma^+$  & 2 & 3&8(--3) & 3(11) & 1&00 & 1&95 & 0&0 & 2169&8 & 13&29 & 5 & c \\
15 & 17 &  939&9574 & 106~387&8 & $I'(5s\sigma)$&$^1\Sigma^+$ & 0 & 2&1(--2) & 1(12) & 1&00 & 2&0365 & 8&8(--5) & 2291&2 & 0&0 & 4 & g, j \\
16 & 18 &  941&1685 & 106~250&9 & $W(3s\sigma)$&$^1\Pi$ & 2 & 3&1(--2) & 1(11) & 1&00 & 1&6224 & $-1$&3(--5) & 1745&1 & $-4$&22 & 1, 6 & g, i, k \\
17 & 19 &  946&2860 & 105~676&3 & $(4d\sigma)$&$^1\Sigma^+$ & 0 & 7&6(--3) & 1(11) & 1&00 & 1&8952 & 1&73(--5) & 2214&2 & 15&15 & 1, 2, 3 & d \\
18 & 20 &  948&3860 & 105~442&3 & $L(4p\pi)$&$^1\Pi$ & 1 & 2&8(--3) & 1(10) & 0&99 & 1&9606 & 1&0(--5) & 2170&5 & 0&0 & & g \\
19 & 21 &  950&0429 & 105~258&4 & $H(4p\sigma)$&$^1\Sigma^+$ & 1 & 2&2(--2) & 1(12) & 1&00 & 1&9436 & 4&4(--5) & 2203&7 & 0&0 & 4 & g, j \\
20 & 22 &  956&2369 & 104~576&6 & $W(3s\sigma)$&$^1\Pi$ & 1 & 1&6(--2) & 7(11) & 1&00 & 1&5696 & 5&8(--5) & 1745&1 & $-4$&22 & 6 & g \\
21 & 24 &  964&3973 & 103~691&7 & $J(4s\sigma)$&$^1\Sigma^+$ & 1 & 2&8(--3) & 3(11) & 1&00 & 1&9202 & 9&0(--6) & 2235&7 & 0&0 & & g \\
22 & 25 &  968&3187 & 103~271&8 & $L(4p\pi)$&$^1\Pi$ & 0 & 1&4(--2) & 2(9)\phantom{0} & 0&96 & 1&9598 & 7&06(--6) & 2170&5 & 0&0 & 7, 8, 9 & g, i, k \\
23 & 26 &  968&8815 & 103~211&8 & $L'(3d\pi)$&$^1\Pi$ & 1 & 1&2(--2) & 2(11) & 1&00 & 1&754 & 1&0(--5) & 2214&2 & 15&15 & 1, 2, 3, 9 & d, k \\
24 & 27 &  970&3588 & 103~054&7 & $K(4p\sigma)$&$^1\Sigma^+$ & 0 & 3&4(--2) & 2(10) & 0&99 & 1&9166 & 5&99(--5) & 2203&7 & 0&0 & 1, 2, 3, 9 & g \\
25 & 28 &  972&6996 & 102~806&7 & $W(3s\sigma)$&$^1\Pi$ & 0 & 1&7(--2) & 1(10) & 0&97 & 1&5664 & 9&7(--5) & 1745&1 & $-4$&22 & 1, 3, 6 & g, i, k \\
26 & 29 &  977&3996 & 102~312&3 & $W'(3s\sigma)$&$^3\Pi$ & 2 & 1&8(--3) & 4(11) & 1&00 & 1&5417 & 8&0(--6)  & 1562&8 & 13&93 & 1, 2, 3 & f, k \\
27 & 30 &  982&5914 & 101~771&7 & $F(3d\sigma)$&$^1\Sigma^+$ & 1 & 4&8(--4) & 3(11) & 1&00 & 1&8506 & 1&4(--5) & 2030&0 & 0&0 & & g \\
28 & 31 &  985&6490 & 101~456&0 & $J(4s\sigma)$&$^1\Sigma^+$ & 0 & 1&5(--2) & 1(12) & 1&00 & 1&9247 & 5&1(--5) & 2235&7 & 0&0 & 4 & g, j  \\
29 & 32 &  989&7952 & 101~031&0 & $G(3d\pi)$&$^1\Pi$ & 0 & 4&6(--4) & 1(11) & 1&00 & 1&9632 & 1&1(--5) & 2214&2 & 15&15 & & d \\
30 & 33 & 1002&5897 &  99~741&7 & $F(3d\sigma)$&$^1\Sigma^+$ & 0 & 7&9(--3) & 3(11) & 1&00 & 1&8077 & 2&2(--4) & 2030&0 & 0&0 & & g \\
31 & 37 & 1051&7134 &  95~082&9 & $E(3p\pi)$&$^1\Pi$ & 1 & 3&6(--3) & 6(9)\phantom{0} & 0&96 & 1&9280 & 6&64(--6) & 2238&8 & 42&51 & 10, 11, 12, 13 & k \\
32 & 38 & 1063&0880 &  94~065&6 & $C(3p\sigma)$&$^1\Sigma^+$ & 1 & 3&0(--3) & 2(9)\phantom{0} & 0&56 & 1&9238 & 6&25(--6) & 2175&9 & 14&76 & 14, 15, 16 & \\
33 & 39 & 1076&0796 &  92~929&9 & $E(3p\pi)$&$^1\Pi$ & 0 & 6&8(--2) & 1(9)\phantom{0} & 0&80 & 1&9527 & 6&34(--6) & 2238&8 & 42&51 & 13, 15, 17, 18 & k \\
\hline
\end{tabular}}}$$
\medskip
{
References: (1) $\lambda_0$ and $\nu_0$ from \citet{ubachs94a}; (2) $A_\el{tot}$ from \citet{ubachs94a}; (3) $\bvu$ and $\dvu$ from \citet{ubachs94a}; (4) $\lambda_0$ and $\nu_0$ from vDB88; (5) $\bvu$ and $\dvu$ from vDB88; (6) $\osc$ and $A_\el{tot}$ from \citet{eidelsberg06a}; (7) $\lambda_0$, $\nu_0$, $\bvu$ and $\dvu$ from \citet{cacciani02a}; (8) $A_\el{tot}$ from \citet{drabbels93b}; (9) $\osc$ from \citet{eidelsberg04a}; (10) $\lambda_0$, $\nu_0$, $\bvu$ and $\dvu$ from \citet{ubachs00a}; (11) $\osc$ from \citet{eidelsberg06a}; (12) $A_\el{tot}$ from \citet{ubachs00a}; (13) $\weu$, $\wexeu$ from \citet{kepa88a}; (14) $\lambda_0$, $\nu_0$, $A_\el{tot}$, $\bvu$ and $\dvu$ from \citet{cacciani01a}; (15) $\osc$ from \citet{federman01a}; (16) $\weu$, $\wexeu$ from \citet{tilford68a}; (17) $\lambda_0$, $\nu_0$, $\bvu$ and $\dvu$ from \citet{cacciani04a}; (18) $A_\el{tot}$ from \citet{cacciani98a}. \\
Notes. 
$^{a}$ $\lambda_0$, $\nu_0$, ID and $v'$ from \citet{eidelsberg92a}, $\osc$, $A_\el{tot}$, $\bvu$ and $\dvu$ from ER90, unless noted otherwise. All $\eta$ are recomputed from the $\osc$ and $A_\el{tot}$. The notation $a(b)$ in this and following tables means~\scit{a}{b}. Many values are rounded off from higher-precision values in the references. \\
$^{b}$~The numbering follows vDB88. Their bands~1 and~2 are split into four and two components. The corresponding ER90{} indices are also given. \\
$^{c}$~$\weu$ and $\wexeu$ from the CO ground state \citep{guelachvili83a}. $\weyeu$ and $\wezeu$ (not listed) are included in the model. \\
$^{d}$~$\weu$ and $\wexeu$ from the \cop{} $X~^2\Sigma^+$~state \citep{haridass00a}. \\
$^{e}$~$\bvu$ and $\dvu$ from the \cop{} $A~^2\Pi$~state \citep{haridass00a}. \\
$^{f}$~$\weu$ and $\wexeu$ from the \cop{} $A~^2\Pi$ state \citep{haridass00a}. $\weyeu$ (not listed) is included in the model. \\
$^{g}$~Vibrational constants derived from the different $\nu_0$ in one of six vibrational series: bands 30$-$27, 28$-$21, 25$-$20--16--8, 24--19, 22--18 or 15--5. \\
$^{h}$~$\bvu$ and $\dvu$ from the \cop{} $X~^2\Sigma^+$ state \citep{haridass00a}. \\
$^{i}$~$A_\el{tot}$ depends on parity and/or rotational level (see \tb{parrota}). $A_\el{tot}$ and $\eta$ are listed here for $J'=0$ and $f$~parity. \\
$^{j}$~$\bvu$ and $\dvu$ computed from the \coo{18} values of ER90. \\
$^{k}$ $\bvu$ and $\dvu$ depend on parity (see \tb{parbc}); values are listed here for $f$ parity.}
\end{table}